Author's response to reviews

Title: Giant aneurysm of the left main coronary artery with fistulous communication to the right atrium

Authors:

Liu Kexiang (kxliu64@hotmail.com)
Xu Rihao (xu6mouse@hotmail.com)
Wang Yong (wangyong.hahaha@163.com)
Zhu Zhicheng (zhu5wolf@hotmail.com)
Li Dan (lidan99@jlu.edu.cn)
Wang Tiance (w19670721@163.com)
Li Bo (adam1911@21cn.com)
Zhang Shudong (zsd_888905@sina.com)

Version: 2
Date: 6 January 2015

Author’s response to reviews:

1# submit a populated CARE checklist.
2# The title page add corresponding author details and complete co-author email addresses.
3# Add an Introduction section.
4# Add a list of abbreviations used in the manuscript.
5# Add authors’ Contributions section at the end of the manuscript.
6# Add an acknowledgement section at the end of the manuscript.